Author's response to reviews

Title: Bone Mineral Density in healthy Iranian population: A Cross sectional study

Authors:

Bagher Larijani (emrc@sina.tums.ac.ir)
Arash Hossein-Nezhad (arash_hsi@yahoo.com)
Alireza Mojtahedi (amojtaheidi@msn.com)
Mohammad Pajouhi (emrc@sina.tums.ac.ir)
Mohammad HASSAN Bastanagh (emrc@sina.tums.ac.ir)
Akbar Soltani (emrc@sina.tums.ac.ir)
Seyed Zahra Mirfeizi (emrc@sina.tums.ac.ir)
Roya Dashti (emrc@sina.tums.ac.ir)

Version: 2  Date: 20 December 2004

Author's response to reviews: see over